Author's response to reviews

Title: A cluster randomised pragmatic trial applying Self-determination theory to type 2 diabetes care in general practice

Authors:

Lise Juul (lise.juul@alm.au.dk)
Helle T. Maindal (htm@alm.au.dk)
Vibeke Zoffmann (vzof@steno.dk)
Morten Frydenberg (morten@biostat.au.dk)
Anneli Sandbaek (anneli.sandbaek@alm.au.dk)

Version: 4 Date: 16 November 2011

Author's response to reviews:

Dear Editor of BMC Family Practice

The following changes have now been made;

• The author list in the manuscript is now identical with the author list in the submission system (Helle Terkildsen Maindal changed to Helle T. Maindal)

• In the abstract, the font in the headings is changed from bold to regular. Now the abstract is identical with the abstract in the submission system

• “The study is conducted according to the Helsinki declaration” is added to the description of the ethical approval, which now is placed in the methods section of the manuscript

• “Study protocol”, “words”, “Figures”, and running title are deleted from the title page

• The authors email addresses are included in the suggested format on the title page

• Titles and figure numbers are excluded from the figure files

• All the figures have been cropped.

Best regards,

Lise Juul